Author’s response to reviews

Title: Towards greater understanding of implementation during systematic reviews of complex healthcare interventions: the Framework for Implementation Transferability Applicability Reporting (FITAR)

Authors:

Susan Baxter (s.k.baxter@sheffield.ac.uk)
Maxine Johnson (m.johnson@sheffield.ac.uk)
Duncan Chambers (d.chambers@sheffield.ac.uk)
Anthea Sutton (a.sutton@sheffield.ac.uk)
Elizabeth Goyder (e.goyder@sheffield.ac.uk)
Andrew Booth (a.booth@sheffield.ac.uk)

Version: 2 Date: 27 Nov 2018

Author’s response to reviews:

Please find attached a document outlining our response to the feedback from the reviewers.